\begin{table}%t2
\caption{General  properties  of 54 events XRRs/XRFs: Time UT of the burst trigger, instrument of the observation,  fluences in  $10^{-7}~{\rm erg~cm}^{-2}$ in the energy range 30--400~\kev\ (S[30, 400]), 2--30~\kev\ (S[2, 30]), 40--700~\kev\ (S[40, 700]) and 2--10~\kev\ (S[2, 10]).}\label{tab2} 
\small%\centerline
{
\begin{tabular}{lccccccc}
\hline\hline
Events&Time UT & Instruments  & $S[30,400]$ & $S[2,30]$ & $S[40,700]$ & $S[2,10]$ & Ref  \\
\hline
\object{XRF 971019}   &         & WFC/BATSE off-line    &             &           &             &         &    (1)\\
\object{XRR 971024}   &         & WFC/BATSE off-line    &             &           &             &         &    (1)\\
\object{XRR 980128}   &         & WFC/BATSE off-line    &             &           &             &         &    (1) \\
\object{XRR 980306}   &         & WFC/BATSE off-line    &             &           &             &         &     (1) \\
\object{XRF 981226}   & 26.41  & GRBM,WFC              &     &   & 4 $\pm$ $1$    &  5.7 $\pm$ $1.0$ & (2) \\
\object{XRF 990520}   & 20.09  & WFC/BATSE off-line    &     &   &  8 $\pm$ $3$(12) &               & (1) \\
\object{XRF 990526}   &        & WFC/BATSE off-line    &             &           &             &         &   (1) \\
\object{XRF 990704}   & 4.73   & GRBM, WFC             &     &   &  10 $\pm$ $1$  & 15.0 $\pm$ 0.8  & (3)  \\
\object{XRR 991106}   &  6.45  & GRBM,WFC              &             &           &    $<$$1.2$          &         &    (4) \\
\object{XRF 000206}   &        & WFC/BATSE off-line    &             &           &             &         &     (1) \\
\object{XRR 000208}   &        & WFC/BATSE off-line    &             &           &             &         &     (1) \\
\object{XRF 000416}   &        & WFC/BATSE off-line    &     &   & 2.9 $\pm$ 0.3(12)             &         & (1) \\
\object{XRF 000615}   &        & GRBM,WFC              &     &   & 9.8 $\pm$ 0.9 &  17 $\pm$ $1$ & (5) \\
\object{XRF 010213}   & 13.53  & FREGATE,WXM           & $\rm 0.69^{+0.58}_{-0.32}$ &  $7.9\pm0.3$  & & & (6)\\
\object{XRF 010225}   &        & FREGATE,WXM                      & $\rm 2.4^{+1.7}_{-0.9}$ &  $3.5\pm0.4$  & & & (6)\\
\object{XRR 010326B}  & 26.36  & FREGATE,WXM           & $\rm 3.2^{+0.9}_{-0.8}$ &  $2.4\pm0.3$  & & & (6)\\
\object{XRR 010613}   & 13.32  & FREGATE,WXM            & $228\pm13$ &  $102\pm7$  & & & (6)\\
\object{XRR 010629}   & 29.52  & FREGATE,WXM           & $29\pm3$ &  $25\pm2$  & & & (6)\\
\object{XRR 010921}   & 21.22  & FREGATE,WXM           & $\rm 113^{+9}_{-8}$ &  $72\pm3$  & & & (6)\\
\object{XRF 011019}   & 19.36  & FREGATE,WXM           & $\rm 1.1^{+1.4}_{-0.7}$ &  $3.0\pm0.6$  & & & (6)\\
\object{XRR 011030}   & 30.27  & WFC                   &    &    &   &  1.2(13)  & (7) \\
\object{XRR 011103}   &   & FREGATE,WXM         & $\rm 6^{+9}_{-3}$ &  $\rm 3.3^{+0.8}_{-0.7}$  & & & (6)\\
\object{XRF 011130}   & 30.26  & FREGATE,WXM           & $\rm 0.98^{+1.17}_{-0.62}$ &  $5.9\pm0.1$  & & & (6)\\
\object{XRR 011211}    & 11.88  & GRBM, WFC             &     &   &  37 $\pm$ $4$  & 11 $\pm$ 1  & (8)  \\
\object{XRF 011212}    & 12.17  & FREGATE, WXM           & $\rm 3.4^{+2.5}_{-1.7}$ &  $4.2\pm0.6$  & & & (6)\\
\object{XRR 020124}     & 24.45  & FREGATE,WXM           & $\rm 61^{+9}_{-8}$ &  $20\pm1$  & & & (6)\\
\object{XRR 020127}     & 27.87  & FREGATE,WXM           & $\rm 21^{+5}_{-4}$ &  $6.7\pm0.5$  & & & (6)\\
\object{XRF 020317}     & 17.76  & FREGATE,WXM           & $\rm 1.3^{+0.9}_{-0.6}$ &  $2.2\pm0.4$  & & & (6)\\
\object{XRR 020410}     & 30.27  & WFC, KONUS(WIND)                   &    &    & $\sim$$290$  &  $>$$47$  & (9)\\
\object{XRF 020427}      & 27.18  & WFC                  &   &   &  $<$$2.9$ &  37 $\pm$ $0.3$  & (10) \\
\object{XRF 020625}      & 25.48  & FREGATE,WXM          & $\rm 0.12^{+0.35}_{-0.11}$ &  $\rm 2.4^{+0.6}_{-0.5}$  & & & (6)\\
\object{XRR 020812}      & 12.45  & FREGATE,WXM           & $\rm 19^{+8}_{-6}$ &  $7.9\pm1.1$  & & & (6)\\
\object{XRR 020819}      & 19.62  & FREGATE,WXM,SXC           & $\rm 63^{+8}_{-9}$ &  $25\pm1$  & & & (6)\\
\object{XRF 020903}      & 3.42 & FREGATE,WXM,SXC        & $\rm 0.16^{+0.44}_{-0.14}$ &  $\rm 0.83^{+0.28}_{-0.24}$  & & & (6)\\
\object{XRR 021004}      & 4.50  & FREGATE,WXM,SXC           & $\rm 18^{+7}_{-5}$ &  $7.7\pm0.7$  & & & (6)\\
\object{XRF 021021}      & 21.78  & FREGATE,WXM           & $\rm 0.62^{+1.07}_{-0.49}$ &  $2.5\pm0.6$  & & & (6)\\
\object{XRF 021104}      & 4.29  & FREGATE,WXM           & $\rm 6.1^{+4.4}_{-2.7}$ &  $10\pm2$  & & & (6)\\
\object{XRR 021112}      & 12.15  & FREGATE,WXM           & $\rm 2.1^{+1.1}_{-0.9}$ &  $1.3\pm0.3$  & & & (6)\\
\object{XRR 021211}       & 11.47  & FREGATE,WXM,SXC           & $\rm 23.71^{+2.03}_{-2.01}$ &  $11.6\pm0.3$  & & & (6)\\
\object{XRR 030115}       & 15.14  & FREGATE,WXM,SXC           & $\rm 15^{+4}_{-3}$ &  $7.9\pm0.6$  & & & (6)\\
\object{XRR 030323}       & 23.92  & FREGATE,WXM,SXC           & $\rm 8.9^{+3.8}_{-3.5}$ &  $\rm 3.4^{+1.3}_{-1.2}$  & & & (6)\\
\object{XRR 030324}       & 24.13  & FREGATE,WXM           & $13\pm3$ &  $\rm 5.5^{+0.4}_{-0.5}$  & & & (6)\\
\object{XRR 030329}       & 29.48  & FREGATE,WXM,SXC           & $\rm 1076^{+13}_{-14}$ &  $553\pm3$  & & & (6)\\
\object{XRF 030416}        & 16.46  & FREGATE off-line,WXM         & $\rm 3.7^{+1.9}_{-1.4}$ &  $9.0\pm0.9$  & & & (6)\\
\object{XRR 030418}        & 18.42  & FREGATE,WXM           & $\rm 17^{+7}_{-5}$ &  $17\pm1$  & & & (6)\\
\object{XRF 030429}        & 29.41  & FREGATE,WXM,SXC           & $\rm 3.8^{+1.4}_{-1.2}$ &  $4.7\pm0.5$  & & & (6)\\
\object{XRF 030528}        & 28.55 & FREGATE,WXM           & $ 56\pm7$ &  $63\pm3$  & & & (6)\\
\object{XRF 030723}         & 23.27  & FREGATE,WXM,SXC           & $\rm 0.38^{+5.56}_{-0.33}$ &  $2.8\pm0.5$  & & & (6)\\
\object{XRR 030725}          & 25.49  & FREGATE,WXM           & $167\pm10$ &  $94\pm2$  & & & (6)\\
\object{XRR 030821}          & 21.23  & FREGATE,WXM           & $\rm 28^{+3}_{-2}$ &  $10\pm0.6$  & & & (6)\\
\object{XRF 030823}          & 23.37  & FREGATE,WXM,SXC           & $13\pm4$ &  $23\pm2$  & & & (6)\\
\object{XRF 030824}          & 24.70  & FREGATE,WXM           & $\rm 5.8^{+2.4}_{-1.9}$ &  $8.9\pm1.1$  & & & (6)\\
\object{XRF 031109B}         & 19.07 & FREGATE,WXM     & & & & & (11) \\
\object{XRR 031220}          & 20.15 & FREGATE,WXM,SXC  & & & & & (11)\\
\hline
\end{tabular}}
\medskip
 (1): \citet{kip03}, (2): \citet{fro00}, (3): \citet{fero01}, (4): \citet{gand99}, (5): \citet{Miao04}, (6): \citet{Sakamoto}, (7): \citet{galli}, (8): \citet{piro04},  (9): \citet{nic04},  (10): \citet{ama04},
(11): http:space.mit = {\tt http:space.mit.edu/HETE/BURST}, (12): fluence between $50{-}300$~keV, (13): fluence between $2{-}28$~keV.
\end{table}